Telehealth and Australian general practice in 2020

A survey exploring patients' perspectives in the Adelaide Hills

Thomas Mathew, Phey Chien Lee, Damian James Ianno, Jill Benson

Background and objective

COVID-19 and Medicare Benefits Schedule rebates have driven the increased use of telehealth in Australian general practice. The aim of this study was to gain an understanding of patients' perspectives towards telehealth in general practice.

Methods

A 10-question online survey was designed and distributed to patients in the Adelaide Hills via direct email invitation, social media posts and flyers between 6 June and 17 July 2020.

Results

A total of 154 responses were obtained, 84% indicating interest in ongoing use of telehealth. Telephone consultations made up 100% of consultations. Six per cent of patients would have preferred video consultations. Seventeen per cent would pay an out-of-pocket fee, while the remaining would only use telehealth if bulk billed. No concerns regarding privacy, technical difficulty or lack of confidence in assessments were expressed.

Discussion

The reception of telephone consultations in general practice has been strongly positive. Key drivers towards this success should be considered by governing bodies if there is to be continued innovation within the area of remote healthcare delivery.

COVID-19 HAS LED TO MANY CHANGES in the population's day-to-day practices, particularly in the healthcare sector. ^{1,2} With the introduction of Medicare Benefits Schedule (MBS) rebates for telehealth in general practice, there has been a significant increase in the use of telehealth throughout Australia. ³ Although there is an established literature on the use of telehealth and its benefits, there is a significant gap in knowledge about the patient's perspective towards this service, especially within the pandemic setting of COVID-19. ⁴⁻⁷

The aim of this online study was to further understand:

- overall patient acceptance of telehealth
- · factors that influence patient acceptance of telehealth
- patients' preferred mode of telehealth
- willingness to pay an out-of-pocket fee for telehealth.

Methods

Study design

A short, anonymous 10-question survey was designed using the SurveyMonkey platform. This platform was chosen because of its user-friendly interface. Data collection took place within the Adelaide Hills region, and respondents aged ≥18 years who had a recent telehealth consultation were eligible to participate. To assess the survey design and format, three doctors and three members of the public were chosen to complete and evaluate the survey. Feedback was obtained on the length, structure, syntax and readability of the survey; through group consultation, a consensus was reached on the final version to be used in the study. Only minor changes to wording were required.

Patients were made aware of the survey and invited to respond using the following methods:

- flyers printed flyers containing the title of the study, survey link and QR code were handed out at 10 participating centres, consisting of four general practice clinics and six pharmacies that made up part of the various centres in the region
- social media posts an open invitation including the study title and survey link was posted onto the social media page of a participating clinic within the region

 direct email link – six participating doctors used standardised introductory statements to obtain permission to collect patient email addresses after each telehealth consultation. These addresses were forwarded to the study team, after which survey invitations

were sent directly to the patients. On accessing the links, all respondents were provided with a one-page plain language summary regarding the study and were made aware of the survey's anonymous and voluntary nature prior to commencing the questionnaire. Respondents were required to fully complete the questionnaire prior to submission. The questionnaire administered is shown in Table 1.

Human Research Ethics approval was obtained through Flinders University, South Australia (Project number: HEL 1934), and no funding was received to undertake this study.

Statistical analysis

Descriptive statistics were used to describe survey responses and are represented as overall count and percentage.

Pearson's Chi-squared tests were used to investigate any associations between demographic factors, employment history, government benefit status, general practice consultation details and preferences regarding telehealth services.

Separate logistic regression analyses were used to model odds ratios (ORs) for associations between the dependent variable and the independent variables.

All statistical comparisons were two-tailed, and statistical significance was set at P < 0.05.

Statistical analysis was performed using commercial statistical software STATA (ver. 16; StataCorp, TX, USA).

Sample size calculations were conducted in STATA/IC 16.1 using the *svysampsi* command for surveys with a dichotomous outcome variable, with a proportion of 0.5 assumed to have the expected outcome, an error rate of 10%, a response rate of 70%, and a 95% confidence interval. This gave an estimated required sample size of 137 subjects.

A proportion of 50% was used as the expected proportion to be in favour of

using telehealth after COVID-19 has resolved, as this represented a balanced estimate of probability for a unique event (COVID-19) for which there were no previous data on expected rate for outcome of interest. A 70% response rate was chosen as it was deemed achievable and also considered ideal as a response rate for surveys in the literature.8 A 10% margin of error was used, as this has been deemed appropriate in the literature.9 Responses were collected until this sample size was met, with an intended timeframe of six weeks. The study was to continue in two-week increments beyond six weeks if the desired sample size was not achieved.

Results

The key variable of interest for this study was the likelihood of using telehealth services after the COVID-19 environment has resolved. Explanatory variables relate to patient demographics, employment history, government benefit status and general practice consultation details. All included variables are summarised and described in Table 1, together with summary statistics.

Data collection took place over a six-week period from 6 June 2020 until 17 July 2020, during which 154 responses were obtained.

A total of 216 direct email links were sent out, and 131 responses were obtained. Forty-four emails were unopened, 10 emails bounced and three respondents opted out. In 28 cases, email invitations were opened but the survey was not attempted. Twenty-three responses in total were obtained via weblink (both via flyers and social media link).

The response rate for direct email links was estimated to be 60.6%.

There were no responses excluded, nor were there any technical issues reported to the study team.

Observing the age demographics of respondents, 23% (the largest proportion) were aged 30–39 years, while on both ends of the age spectrum (18–29 years and >75 years) the proportions of respondents were observed to be lower.

Employment status differed when compared with state-wide data, where

employment in South Australia is noted at 56% (compared with 60% in the present study) and unemployment at 8.8% (compared with 6% in the present study) of the population. ¹⁰ This reflects the trend that has been observed in previous census data and could be related to the growing establishments in the Adelaide Hills region. ¹¹

All respondents indicated that their telehealth consultation was undertaken via telephone, and only 6% would have preferred video consultation. Telehealth usage prior to COVID-19 was small, with only 11% indicating they had previously used telehealth.

When looking at the types of consultations, 'test results' made up the largest group (30%), followed by 'repeat prescriptions' (23%) and 'management of pre-existing illnesses' (10%).

Of these consultations, 92% were able to be addressed by the treating doctor at the time of the telehealth consultation, with only 6% (10 respondents) asked to rebook for a face-to-face consultation. None of the respondents indicated technical difficulty or low confidence in the assessment done via telehealth, with only one respondent indicating they were unhappy with the management plan/advice provided.

Overall sentiment towards telehealth services was positive, with 84% indicating they would be keen to use this service after the COVID-19 pandemic has resolved. Seventeen per cent indicated they would be willing to pay standard out-of-pocket fees, while 67% would only use telehealth services if bulk billed.

Of the 17% respondents who were not keen to use telehealth services, 12% preferred face-to-face consultations, while 5% stated 'other reasons'. One respondent stated, 'Every consultation is for different reasons; should an examination be needed, it would be a great waste of time to have to wait to book another appointment'. None of the respondents declined ongoing use of telehealth because of privacy concerns or lack of familiarity with technology, although it must be remembered that telephone consultations made up 100% of the mode of delivery.

To further assess the overall acceptance of telehealth (Question 9 of survey,

Table 1. Summary o	f responses ((n = 154)
--------------------	---------------	-----------

Table 1. Summary of responses (n		Number
1 Disease colors vous current are group	Percentage	Number
1. Please select your current age group	15%	22
18-29		23
30-39	23%	36
40-49	21%	32
50-59	18%	28
60-75	14%	21
>75 ————————————————————————————————————	9%	14
2. Please select your gender		
Male	32%	49
Female	68%	105
Other	0%	0
3. What is your current employment st	tatus?	
Student	6%	9
Employed	60%	93
Unemployed	3%	5
Unemployed due to COVID-19	3%	4
Disabled, unable to work	0%	0
Stay-at-home parent	8%	13
Retired	19%	30
4. Are you a Department of Veterans' A Health care card or Pensioner card hol		
Yes	31%	47
No	69%	107
5. Have you used telehealth services prior to COVID-19?		
Yes	11%	17
No	89%	137
6. Did you have a phone or video consu your recent doctor's appointment?	ult at	
Phone consult	100%	154
Video consult	0%	0
7. What was the reason for your recent	consult?	
Repeat prescription	23%	36
Referral	8%	13
Medical certificate	5%	7

1	Percentage	Number
7. What was the reason for your recent consult? Cont'd		
Test results	30%	46
New medical illness	9%	14
Management of pre-existing medical condition	10%	16
Mental health consult	4%	6
Pregnancy care	1%	2
General medical advice	8%	13
Prefer not to answer	1%	
8. Was your doctor able to address the reason for your consult via telehealth?		
Yes	92%	142
No, I was asked to rebook for a face-to- face consultation	6%	10
No, I was not confident with the assessment done via telehealth	0%	(
No, I was not happy with the management plan/advice provided	1%	
No, I experienced technical difficulty	0%	(
No, other reasons (please specify)	1%	
9. If telehealth services are still in place, are you likely to use these services after COVID-19 has resolved?		
Yes, and willing to pay standard out-of- pocket fees for the service	17%	26
Yes, only if bulk billed	67%	103
No, I prefer face-to-face consultations	12%	18
No, I am concerned regarding privacy issues	0%	(
No, I am not tech savvy	0%	(
No, other reasons (please specify)	5%	
	stion,	
would you prefer phone or video consults?	79%	12
10. If you answered yes to the previous que: would you prefer phone or video consults? Phone consult Video consult	79% 5%	12

'If telehealth services are still in place, are you likely to use these services after COVID-19 has resolved?') a collapsing of answer categories was used to allow for two distinct methods of statistical analysis to be performed. The survey originally contained six answer categories:

- 1. Yes, and willing to pay standard out-of-pocket fees for the service
- 2. Yes, only if bulk billed
- 3. No, I prefer face-to-face consultations
- 4. No, I am concerned regarding privacy issues
- 5. No, I am not tech savvy
- 6. No, other reasons [into three categories].

The first method of analysis involved collapsing the possible answers into three broad categories, these being:

- 1. Yes, and willing to pay standard out-of-pocket fees for the service
- 2. Yes, only if bulk billed
- 3. No [options 3-6 combined].

This facilitated analysis in the form of a Pearson's Chi-squared test (Table 2).

Within the three-group analysis, there were no significant associations found.

The second method of analysis involved collapsing the possible answers into two broad categories, these being:

- 1. Yes [options 1–2 combined]
- 2. No [options 3-6 combined].

With the dichotomised method described above, both Pearson's Chi-squared test and logistic regression were used as statistical methods, and corresponding effects sizes reported as ORs and 95% confidence intervals (CI). The data can be found in Appendix 1, available online only.

No significant relationship was found for the likelihood to attend telehealth for all explanatory variables on Chi-squared analysis and logistic regression analysis.

To assess if there was a difference between responses collected via direct email link and those collected via weblink, a sensitivity analysis was performed. Chi-squared testing revealed that the groups were overall well balanced except for their reason for consultation. When compared with the weblink group, the direct email link group had many more individuals with test results (34%, compared with 9%) and general medical advice (10%, compared with 0%) as a reason for consultation. The weblink group had many more individuals with new medical illness (17%, compared with 8%) and management of a pre-existing condition (35%, compared with 6%) as a reason for consultation.

While the reason for this discrepancy is unclear, a possible explanation could be that flyers with weblinks were handed out at clinics/pharmacies, and the likely reason for patients to present to these centres after a telehealth consultation would be to pick up a prescription or medication to manage a new medical illness/pre-existing condition. Most respondents who had a telehealth consultation for test results or general medical advice would not have had an opportunity to receive the weblink flyers, as those reasons for consultation would generally have been addressed without needing to present to a clinic or pharmacy afterwards.

Discussion

The responses obtained show that the use of telehealth in general practice has been well received, with 84% of respondents indicating they would be keen to use this service after the pandemic has resolved.

Although the number of respondents appears to be smaller in the age groups of 18–29 years and >75 years, the former is likely due to the low number of general practice presentations within this demographic, while in those aged >75 years, computer literacy in accessing the online-based survey may have acted as a barrier.¹²

Easing COVID-19 restrictions one month prior to the data collection period meant that rather than telehealth being the default, patients were more likely to have had the option to choose the most suitable form of consultation (telehealth or face to face) depending on their reason for the consultation. Considering that 92% of consultations were successfully addressed using telehealth, patients were likely making the appropriate choice when deciding between booking their general practice appointments via telehealth or face to face, leading to the overall satisfactory experience of telehealth.

Telephone consultations were the only mode of telehealth delivery within the studied population, and this is consistent with national data published by Snoswell et al. 15 This could be attributed to the rapid shift towards telehealth due to COVID-19, as practices may not have had the infrastructure or time to incorporate video services into their day-to-day practice. As familiarity with Zoom and other video platforms grow, more patients and practices may begin

Table 2. Association between likelihood to attend telehealth and explanatory variables (three-group analysis)

Variable	Three-group analysis	
	χ²	<i>P</i> value
Age (years)	13.4397	0.200
Gender	1.6273	0.443
Employment status	8.6194	0.569
Department of Veterans' Affairs/Health care card/Pensioner card status	4.9654	0.084
Used telehealth services prior to COVID-19	4.6271	0.099
Reason for recent consultation	20.2345	0.320
Reason addressed at telehealth appointment	0.7587	0.684

to use videoconferencing for telehealth because of the added value that it may bring to a consultation, including the ability to perform certain examinations as well as improved rapport through non-verbal cues, which could also reduce misunderstandings between doctors and patients. ^{16,17} Conversely, concerns regarding privacy and technical complications may also start to emerge with an increase in use of video consultations.

MBS rebates introduced by the Australian Government were a catalyst towards the success of telehealth in general practice. The mandatory bulk-billing clause for patients at risk of COVID-19 further meant that many patients had the opportunity to trial this service without financial risk.3 Considering the large proportion of respondents indicating their ongoing use of telehealth only if the service is bulk billed, there may be a significant drop in its use should the MBS item numbers cease at the end of this pandemic. However, if consumers are given the option of receiving a service with no out-of-pocket cost, this is likely to be the choice for many. The number of patients who are keen to pay out-ofpocket fees may grow over time because of the convenience telehealth provides. As stated by one of the respondents, 'It's a much better thing waiting at home rather than the waiting room for close to an hour most times'.

Limitations

As this study was assessing the population at a local level, it may not be representative of responses Australia wide; therefore, external validity and generalisability remains a limitation of this study. However, since commencement, there have been additional data published within Australia that have produced comparable findings to those described in the present study. 15,16,18

The potential of recruitment bias must be considered, as it may have been difficult for participating doctors to discuss the opportunity to contribute to this study if they had a challenging telehealth consultation with the patient. As a result of the larger number of responses collected via direct email link, the likelihood of

this bias affecting the results could be higher. Elimination of this bias would require participant recruitment to be done by a party independent of the patient's telehealth consultation, and this could be considered in future telehealth studies.

The term 'telehealth' encompasses both video and telephone consultations. However, considering the lack of video consultations, the results of this study are only representative of the user experience of telephone consultations.

Finally, the findings from this study are hypothesis generating and may provide valuable data for future studies looking to assess the applicability, use and perceived importance of telehealth. Formal conclusions would require a larger sample size that was beyond the scope of this study and could be addressed in the future with a larger multicentre/state-wide study with the aforementioned recommendations.

Conclusion

On the basis of the responses provided, it can be concluded that the overall acceptance rate of patients towards telephone consultations in general practice has been overwhelmingly positive. If there is to be continuous growth and innovation within the area of remote healthcare delivery, key factors including the ongoing provision of MBS rebates should strongly be considered and advocated by governing bodies. 19 As we progress through the various reforms, healthcare practitioners should be encouraged to engage in continued education and research to enable effective integration of telehealth services within the current model of primary care. Finally, further studies in the area of video consultations should be considered once their use in general practice is more largely established.

Authors

Thomas Mathew MBBS (Hons), General Practice Registrar, GPEx, Unley, SA

Phey Chien Lee MBBS, Resident Medical Officer, Central Adelaide Local Health Network, Adelaide, SA Damian James Ianno BBiom (Hons), MD, Emergency Medicine Registrar, Eastern Health, Melbourne, Vic Jill Benson AM, MBBS, PhD, FRACGP (Hon), Medical Educator, GPEx, Unley, SA; Prideaux Centre, Flinders University, Adelaide, SA Competing interest: None.

Funding: None.

Provenance and peer review: Not commissioned, externally peer reviewed.

Correspondence to:

drthomasmath@gmail.com

References

- Fisk M, Livingstone A, Pit SW. Telehealth in the context of COVID-19: Changing perspectives in Australia, the United Kingdom, and the United States. J Med Internet Res 2020;22(6):e19264. doi: 10.2196/19264.
- Desborough J, Hall Dykgraaf S, de Toca L, et al. Australia's national COVID-19 primary care response. Med J Aust 2020;213(3):104-06.e1. doi: 10.5694/mja2.50693.
- Australian Government Department of Health. COVID-19 temporary MBS telehealth services. Canberra, ACT: MBS Online, 2020. Available at www.mbsonline.gov.au/internet/mbsonline/ publishing.nsf/Content/Factsheet-TempBB [Accessed 24 July 2020].
- Royal Australian College of General Practitioners.
 On demand telehealth services. East Melbourne, Vic: RACGP, 2020. Available at www.racgp.org.au/advocacy/position-statements/view-all-position-statements/health-systems-and-environmental/on-demand-telehealth-services [Accessed 1 May 2020].
- Kruse CS, Krowski N, Rodriguez B, Tran L, Vela J, Brooks M. Telehealth and patient satisfaction: A systematic review and narrative analysis. BMJ Open 2017;7(8):e016242. doi: 10.1136/bmjopen-2017-016242.
- McPhee E. Telehealth: The general practice perspective. Aust Fam Physician 2014;43(12):826–27.
- Bradford NK, Caffery LJ, Smith AC. Awareness, experiences and perceptions of telehealth in a rural Queensland community. BMC Health Serv Res 2015;15:427. doi: 10.1186/s12913-015-1094-7.
- Corner B, Lemonde M. Survey techniques for nursing studies. Can Oncol Nurs J 2019;29(1):58–60.
- Nayak BK. Understanding the relevance of sample size calculation. Indian J Ophthalmol 2010;58(6):469-70. doi: 10.4103/0301-4738.71673.
- Australian Bureau of Statistics. Labour force, Australia. Canberra, ACT: ABS, 2020. Available at www.abs.gov.au/statistics/labour/employmentand-unemployment/labour-force-australia/ oct-2020 [Accessed 12 December 2020].
- Australian Bureau of Statistics. Region summary: Adelaide Hills (DC). Canberra, ACT: ABS, 2020. Available at https://dbr.abs.gov.au/region. html?lyr=lga&rgn=40120 [Accessed 20 August 2021].
- National Health Performance Authority. Healthy communities: Frequent GP attenders and their use of health services in 2012–13.
 Sydney, NSW: NHPA, 2015 Available at www.aihw.gov.au/getmedia/570c83c4bfb4-48b5-807d-935a1db34a1e/ aihw-mhc-nhpa-11-frequent-gp-attenders-anduse-of-health-services-2012-13-report-March-2015. pdf.aspx [Accessed 12 December 2020].
- Government of South Australia. Recovery from COVID-19. Adelaide, SA: Government of South Australia, 2020. Available at www.covid-19.sa.gov. au/recovery [Accessed 9 July 2020].
- Royal Australian College of General Practitioners. RACGP survey reveals strong take up of telehealth but face to face consultations still

- available. East Melbourne, Vic: RACGP, 7 May 2020. Available at www.racgp.org.au/gp-news/media-releases/2020-media-releases/may-2020/racgp-survey-reveals-strong-take-up-of-telehealth [Accessed 26 July 2020].
- Snoswell CL, Caffery LJ, Haydon HM, Thomas EE, Smith AC. Telehealth uptake in general practice as a result of the coronavirus (COVID-19) pandemic. Aust Health Rev 2020;44(5):737-40. doi: 10.1071/ AH20183.
- HotDoc. 7 Findings from HotDoc's Telehealth Patient Survey 2020. Melbourne, Vic: HotDoc, 2020. Available at practices.hotdoc.com.au/blog/ key-findings-telehealth-survey-2020 [Accessed 4 October 2020].
- 17. Avant Media. Australia's largest telehealth survey: What did you tell us? Sydney, NSW: Avant Mutual, 2020. Available at www.avant.org.au/news/

- australias-largest-telehealth-survey [Accessed 4 October 2020].
- HealthEngine. Uptake of telehealth: HeathEngine findings & insights. Perth, WA: Health Engine, 2020. Available at hepress.wpengine.com/ wp-content/uploads/2020/07/Uptake-of-Telehealth-HealthEngine-Insights.pdf [Accessed 4 October 2020].
- 19. Bollen C, Haddock R. Providing telehealth in general practice during COVID-19 and beyond. Deakin, ACT: Deeble Institute for Health Policy Research, 2020;10. Available at ahha.asn.au/sites/default/files/docs/policy-issue/perspectives_brief_no_10_providing_telehealth_in_general_practice_0. pdf [Accessed 4 October 2020].

correspondence ajgp@racgp.org.au